Author's response to reviews

Title: Cholesteatoma of the concha bullosa: a case report

Authors:

Ibrahim Cukurova (cukurova@turk.net)
Erhan Demirhan (demirhanerhan@yahoo.com)
Ilker B Arslan (ilkerburakarslan@yahoo.com)
Suheyla Cumhurcu (scumhurcu@yahoo.com)

Version: 2 Date: 26 August 2010

Author's response to reviews:

- The author list in the manuscript has been corrected both in style and order. Surname of the one author has corrected. (“Suheyla Cumhurcu” is corrected as “Suheyla Cumurcu”)

- The ethnicity of the patient was stated both in the abstract and the manuscript.

- Smaller sized images have been uploaded but these are the images with highest resolution that we have.

- Name of the Figure 3a, 3b were changed as figure3 and 4, respectively. It has also corrected in the manuscript.

- References were also corrected and square brackets were used in the text.

- Style of the Author’s contribution section was changed as a paragraph.